\nopagebreak
\begin{thebibliography}{99}
\frenchspacing
\newcommand{\epj}[3]{{\sl Eur. Phys. J.} {\bf #1} (19#2) #3}
\newcommand{\zp}[3]{{\sl Z. Phys.} {\bf #1} (19#2) #3}
\newcommand{\np}[3]{{\sl Nucl. Phys.} {\bf #1} (19#2) #3}
\newcommand{\phm}[3]{{\sl Phil. Mag.} {\bf #1} (19#2) #3}
\newcommand{\pl}[3]{{\sl Phys. Lett.} {\bf #1} (19#2) #3}
\newcommand{\pr}[3]{{\sl Phys. Rev.} {\bf #1} (19#2) #3}
\newcommand{\prep}[3]{{\sl Phys.\ Rep.} {\bf #1} (19#2) #3}
\newcommand{\prl}[3]{{\sl Phys. Rev. Lett.} {\bf #1} (19#2) #3}
\newcommand{\prs}[3]{{\sl Proc. Roy. Soc.} {\bf #1} (19#2) #3}
\newcommand{\fp}[3]{{\sl Fortschr. Phys.} {\bf #1} (19#2) #3}
\newcommand{\cpc}[3]{{\sl Comput. Phys. Commun.} {\bf #1} (19#2) #3}
\newcommand{\ijmp}[3]{{\sl Int. J. Mod. Phys.} {\bf #1} (19#2) #3}
\newcommand{\nim}[3]{{\sl Nucl. Instr. Meth.} {\bf #1} (19#2) #3}
\newcommand{\nc}[3]{{\sl Nuovo Cimento} {\bf #1} (19#2) #3}
\newcommand{\vj}[4]{{\sl #1} {\bf #2} (19#3) #4}
\newcommand{\jcp}[3]{{\sl J. Comp. Phys.} {\bf #1} (19#2) #3}

%\cite{Aad:2012awa}
\bibitem{Aad:2012awa}
  G.~Aad {\it et al.}  [ATLAS Collaboration],
  %``Measurement of $ZZ$ production in $pp$ collisions at $\sqrt{s}=7$ TeV and limits on anomalous $ZZZ$ and $ZZ\gamma$ couplings with the ATLAS detector,''
  JHEP {\bf 1303} (2013) 128
  [arXiv:1211.6096 [hep-ex]].
  %%CITATION = ARXIV:1211.6096;%%
  %18 citations counted in INSPIRE as of 25 Sep 2013
%\cite{ATLAS:2012mec}
\bibitem{ATLAS:2012mec}
  G.~Aad {\it et al.}  [ATLAS Collaboration],
  %``Measurement of $W^+W^-$ production in $pp$ collisions at $\sqrt{s}=7$ TeV with the ATLAS detector and limits on anomalous $WWZ$ and $WW\gamma$ couplings,''
  Phys.\ Rev.\ D {\bf 87} (2013) 112001
  [arXiv:1210.2979 [hep-ex]].
  %%CITATION = ARXIV:1210.2979;%%
  %30 citations counted in INSPIRE as of 25 Sep 2013
%\cite{Aad:2012twa}
\bibitem{Aad:2012twa}
  G.~Aad {\it et al.}  [ATLAS Collaboration],
  %``Measurement of $WZ$ production in proton-proton collisions at $\sqrt{s}=7$ TeV with the ATLAS detector,''
  Eur.\ Phys.\ J.\ C {\bf 72} (2012) 2173
  [arXiv:1208.1390 [hep-ex]].
  %%CITATION = ARXIV:1208.1390;%%
  %19 citations counted in INSPIRE as of 25 Sep 2013
%\cite{Chatrchyan:2013yaa}
\bibitem{Chatrchyan:2013yaa}
  S.~Chatrchyan {\it et al.}  [CMS Collaboration],
  %``Measurement of the W+W- cross section in pp collisions at sqrt(s) = 7 TeV and limits on anomalous WW gamma and WWZ couplings,''
  arXiv:1306.1126 [hep-ex].
  %%CITATION = ARXIV:1306.1126;%%
  %5 citations counted in INSPIRE as of 25 Sep 2013
%\cite{Chatrchyan:2013oev}
\bibitem{Chatrchyan:2013oev}
  S.~Chatrchyan {\it et al.}  [CMS Collaboration],
  %``Measurement of W+W- and ZZ production cross sections in pp collisions at sqrt(s) = 8 TeV,''
  Phys.\ Lett.\ B {\bf 721} (2013) 190
  [arXiv:1301.4698 [hep-ex]].
  %%CITATION = ARXIV:1301.4698;%%
  %12 citations counted in INSPIRE as of 25 Sep 2013
%\cite{Chatrchyan:2012sga}
\bibitem{Chatrchyan:2012sga}
  S.~Chatrchyan {\it et al.}  [CMS Collaboration],
  %``Measurement of the $ZZ$ production cross section and search for anomalous couplings in 2 l2l ' final states in $pp$ collisions at $\sqrt{s}=7$ TeV,''
  JHEP {\bf 1301} (2013) 063
  [arXiv:1211.4890 [hep-ex]].
  %%CITATION = ARXIV:1211.4890;%%
  %12 citations counted in INSPIRE as of 25 Sep 2013
%\cite{Bierweiler:2012kw}
\bibitem{Bierweiler:2012kw}
  A.~Bierweiler, T.~Kasprzik, H.~K\"uhn and S.~Uccirati,
  %``Electroweak corrections to W-boson pair production at the LHC,''
  JHEP {\bf 1211} (2012) 093
  [arXiv:1208.3147 [hep-ph]].
  %%CITATION = ARXIV:1208.3147;%%
  %5 citations counted in INSPIRE as of 31 Jul 2013
%\cite{Bierweiler:2013dja}
\bibitem{Bierweiler:2013dja}
  A.~Bierweiler, T.~Kasprzik and J.~H.~Kühn,
  %``Vector-boson pair production at the LHC to $\mathcal{O}(\alpha^3)$ accuracy,''
  arXiv:1305.5402 [hep-ph].
  %%CITATION = ARXIV:1305.5402;%%
  %3 citations counted in INSPIRE as of 31 Jul 2013

%\cite{Gehrmann:2013cxs}
\bibitem{Gehrmann:2013cxs}
  T.~Gehrmann, L.~Tancredi and E.~Weihs,
  %``Two-loop master integrals for $q \bar{q} \to VV$: the planar topologies,''
  JHEP {\bf 1308} (2013) 070
  [arXiv:1306.6344 [hep-ph]].
  %%CITATION = ARXIV:1306.6344;%%
  %6 citations counted in INSPIRE as of 23 Sep 2013
%\cite{Chachamis:2008yb}
\bibitem{Chachamis:2008yb}
  G.~Chachamis, M.~Czakon and D.~Eiras,
  %``W Pair Production at the LHC. I. Two-loop Corrections in the High Energy Limit,''
  JHEP {\bf 0812} (2008) 003
  [arXiv:0802.4028 [hep-ph]].
  %%CITATION = ARXIV:0802.4028;%%
  %20 citations counted in INSPIRE as of 23 Sep 2013

%\cite{Bern:2001df}
\bibitem{Bern:2001df}
  Z.~Bern, A.~De Freitas and L.~J.~Dixon,
  %``Two loop amplitudes for gluon fusion into two photons,''
  JHEP {\bf 0109} (2001) 037
  [hep-ph/0109078].
  %%CITATION = HEP-PH/0109078;%%
  %110 citations counted in INSPIRE as of 24 Sep 2013
%\cite{Catani:2011qz}
\bibitem{Catani:2011qz}
  S.~Catani, L.~Cieri, D.~de Florian, G.~Ferrera and M.~Grazzini,
  %``Diphoton production at hadron colliders: a fully-differential QCD calculation at NNLO,''
  Phys.\ Rev.\ Lett.\  {\bf 108} (2012) 072001
  [arXiv:1110.2375 [hep-ph]].
  %%CITATION = ARXIV:1110.2375;%%
  %59 citations counted in INSPIRE as of 24 Sep 2013
%\cite{Grazzini:2013bna}
\bibitem{Grazzini:2013bna}
  M.~Grazzini, S.~Kallweit, D.~Rathlev and A.~Torre,
  %``$Z\gamma$ production at hadron colliders in NNLO QCD,''
  arXiv:1309.7000 [hep-ph].
  %%CITATION = ARXIV:1309.7000;%%
%\cite{Campanario:2012fk}
\bibitem{Campanario:2012fk}
  F.~Campanario and S.~Sapeta,
  %``WZ production beyond NLO for high-pT observables,''
  Phys.\ Lett.\ B {\bf 718} (2012) 100
  [arXiv:1209.4595 [hep-ph]].
  %%CITATION = ARXIV:1209.4595;%%
  %9 citations counted in INSPIRE as of 02 Aug 2013
%\cite{Campanario:2013wta}
\bibitem{Campanario:2013wta}
  F.~Campanario, M.~Rauch and S.~Sapeta,
  %``WW production at high transverse momenta beyond NLO,''
  arXiv:1309.7293 [hep-ph].
  %%CITATION = ARXIV:1309.7293;%%
%\cite{Dawson:2013lya}
\bibitem{Dawson:2013lya}
  S.~Dawson, I.~M.~Lewis and M.~Zeng,
  %``Threshold Resummed and Approximate NNLO results for W+W- Pair Production at the LHC,''
  arXiv:1307.3249 [hep-ph].
  %%CITATION = ARXIV:1307.3249;%%
  %2 citations counted in INSPIRE as of 23 Sep 2013
%\cite{Accomando:2005xp}
\bibitem{Accomando:2005xp}
  E.~Accomando and A.~Kaiser,
  %``Electroweak corrections and anomalous triple gauge-boson couplings in $W^{+} W^{-}$ and $W^\pm Z$ production at the LHC,''
  Phys.\ Rev.\ D {\bf 73} (2006) 093006
  [hep-ph/0511088].
  %%CITATION = HEP-PH/0511088;%%
  %17 citations counted in INSPIRE as of 23 Sep 2013
%\cite{Accomando:2004de}
\bibitem{Accomando:2004de}
  E.~Accomando, A.~Denner and A.~Kaiser,
  %``Logarithmic electroweak corrections to gauge-boson pair production at the LHC,''
  Nucl.\ Phys.\ B {\bf 706} (2005) 325
  [hep-ph/0409247].
%\cite{Billoni:2013aba}
\bibitem{Billoni:2013aba}
  M.~Billoni, S.~Dittmaier, B.~J\"ager and C.~Speckner,
  %``Next-to-leading order electroweak corrections to pp -> W+W- -> 4 leptons at the LHC in double-pole approximation,''
  arXiv:1310.1564 [hep-ph].
  %%CITATION = ARXIV:1310.1564;%%

%\cite{Kuhn:2011mh}
\bibitem{Kuhn:2011mh}
  J.~H.~K\"uhn, F.~Metzler, A.~A.~Penin and S.~Uccirati,
  %``Next-to-Next-to-Leading Electroweak Logarithms for W-Pair Production at LHC,''
  JHEP {\bf 1106} (2011) 143
  [arXiv:1101.2563 [hep-ph]].
  %%CITATION = ARXIV:1101.2563;%%
  %5 citations counted in INSPIRE as of 23 Sep 2013


%\cite{Baglio:2013toa}
\bibitem{Baglio:2013toa}
  J.~Baglio, L.~D.~Ninh and M.~M.~Weber,
  %``Massive gauge boson pair production at the LHC: a next-to-leading order story,''
  arXiv:1307.4331 [hep-ph].
  %%CITATION = ARXIV:1307.4331;%%
  %4 citations counted in INSPIRE as of 23 Sep 2013

%\cite{Chachamis:2013aya}
\bibitem{Chachamis:2013aya}
  G.~Chachamis,
  %``NNLO W Pair Production at the LHC: Status Report,''
  arXiv:1312.4220 [hep-ph].
  %%CITATION = ARXIV:1312.4220;%%

%\cite{Boughezal:2013cwa}
\bibitem{Boughezal:2013cwa}
  R.~Boughezal, Y.~Li and F.~Petriello,
  %``Disentangling radiative corrections using high-mass Drell-Yan at the LHC,''
  arXiv:1312.3972 [hep-ph];
  %%CITATION = ARXIV:1312.3972;%%
%\cite{Kilgore:2011pa}
%\bibitem{Kilgore:2011pa}
  W.~B.~Kilgore, C.~Sturm,
  %``Two-Loop Virtual Corrections to Drell-Yan Production at order $\alpha_s
%\alpha^3$,''
   [arXiv:1107.4798 [hep-ph]];
%\cite{Kotikov:2007vr}
%\bibitem{Kotikov:2007vr}
  A.~Kotikov, J.~H.~K\"uhn, O.~Veretin,
  %``Two-Loop Formfactors in Theories with Mass Gap and Z-Boson Production,''
  Nucl.\ Phys.\  {\bf B788 } (2008)  47-62.
  [hep-ph/0703013 [HEP-PH]];
%\bibitem{Cao:2004yy}
  Q.~H.~Cao and C.~P.~Yuan,
  %``Combined effect of QCD resummation and QED radiative correction to W  boson
  %observables at the Tevatron,''
  Phys.\ Rev.\ Lett.\  {\bf 93} (2004) 042001
  [hep-ph/0401026];
  %%CITATION = PRLTA,93,042001;%%
  B.~F.~L.~Ward, C.~Glosser, S.~Jadach and S.~A.~Yost,
  %``Threshold corrections in precision LHC physics: QED x QCD,''
  Int.\ J.\ Mod.\ Phys.\  A {\bf 20} (2005) 3735
  [hep-ph/0411047];
  %%CITATION = IMPAE,A20,3735;%%
  B.~F.~L.~Ward and S.~A.~Yost,
  %``QED x QCD Resummation and Shower/ME Matching for LHC Physics,''
  Acta Phys.\ Polon.\  B {\bf 38} (2007) 2395
  [arXiv:0704.0294 [hep-ph]];
  %%CITATION = APPOA,B38,2395;%%
%  G.~Montagna, talk given at ``Loopfest VI: Radiative Corrections for the LHC
%and ILC'', Fermilab, 2007.
%\cite{Vicini:2007zz}
%\bibitem{Vicini:2007zz}
  A.~Vicini {\it et al.},
  %``Drell-Yan processes at the LHC,''
  PoS {\bf RADCOR2007} (2007) 013;
  %%CITATION = POSCI,RADCOR2007,013;%%
%\cite{Balossini:2009sa}
%\bibitem{Balossini:2009sa}
  G.~Balossini, G.~Montagna, C.~M.~Carloni Calame, M.~Moretti, O.~Nicrosini, F.~Piccinini, M.~Treccani, A.~Vicini,
  %``Combination of electroweak and QCD corrections to single W production at the Fermilab Tevatron and the CERN LHC,''
  JHEP {\bf 1001 } (2010)  013.
  [arXiv:0907.0276 [hep-ph]];
%\cite{Balossini:2008zz}
%\bibitem{Balossini:2008zz}
  G.~Balossini {\it et al.},
  %``Drell-Yan processes at hadron colliders,''
  Nuovo Cim.\  {\bf 123B} (2008) 741.
  %%CITATION = NUCIA,123B,741;%%

%\cite{Bahr:2008pv}
\bibitem{Bahr:2008pv}
  M.~B{\"a}hr, S.~Gieseke, M.~A.~Gigg, D.~Grellscheid, K.~Hamilton, O.~Latunde-Dada, S.~Pl{\"a}tzer and P.~Richardson {\it et al.},
  %``Herwig++ Physics and Manual,''
  Eur.\ Phys.\ J.\ C {\bf 58} (2008) 639
  [arXiv:0803.0883 [hep-ph]].
  %%CITATION = ARXIV:0803.0883;%%
  %626 citations counted in INSPIRE as of 21 Oct 2013

%\cite{Arnold:2012fq}
\bibitem{Arnold:2012fq}
  K.~Arnold, L.~d'Errico, S.~Gieseke, D.~Grellscheid, K.~Hamilton, A.~Papaefstathiou, S.~Pl{\"a}tzer and P.~Richardson {\it et al.},
  %``Herwig++ 2.6 Release Note,''
  arXiv:1205.4902 [hep-ph].
  %%CITATION = ARXIV:1205.4902;%%
  %22 citations counted in INSPIRE as of 21 Oct 2013
%\cite{Martin:2004dh}
\bibitem{Martin:2004dh}
  A.~D.~Martin, R.~G.~Roberts, W.~J.~Stirling, R.~S.~Thorne,
  %``Parton distributions incorporating QED contributions,''
  Eur.\ Phys.\ J.\  {\bf C39 } (2005)  155-161.
  [hep-ph/0411040].
%\cite{Ball:2013hta}
\bibitem{Ball:2013hta}
  R.~D.~Ball {\it et al.}  [ The NNPDF Collaboration],
  %``Parton distributions with QED corrections,''
  arXiv:1308.0598 [hep-ph].
  %%CITATION = ARXIV:1308.0598;%%
  %2 citations counted in INSPIRE as of 23 Sep 2013
%\cite{Dittmaier:1999mb}
\bibitem{Dittmaier:1999mb}
  S.~Dittmaier,
  %``A General approach to photon radiation off fermions,''
  Nucl.\ Phys.\ B {\bf 565} (2000) 69
  [hep-ph/9904440].
  %%CITATION = HEP-PH/9904440;%%
  %142 citations counted in INSPIRE as of 31 Jul 2013
%\cite{Hamilton:2010mb}
\bibitem{Hamilton:2010mb}
  K.~Hamilton,
  %``A positive-weight next-to-leading order simulation of weak boson pair production,''
  JHEP {\bf 1101} (2011) 009
  [arXiv:1009.5391 [hep-ph]].
  %%CITATION = ARXIV:1009.5391;%%
  %18 citations counted in INSPIRE as of 21 Oct 2013


%\cite{Lai:2010vv}
\bibitem{Lai:2010vv}
  H.~-L.~Lai, M.~Guzzi, J.~Huston, Z.~Li, P.~M.~Nadolsky, J.~Pumplin and C.~-P.~Yuan,
  %``New parton distributions for collider physics,''
  Phys.\ Rev.\ D {\bf 82} (2010) 074024
  [arXiv:1007.2241 [hep-ph]].
  %%CITATION = ARXIV:1007.2241;%%
  %599 citations counted in INSPIRE as of 02 Aug 2013
%\cite{Whalley:2005nh}
\bibitem{Whalley:2005nh}
  M.~R.~Whalley, D.~Bourilkov and R.~C.~Group, in {\sl HERA and the
    LHC}, eds.\ A. de Roeck and H. Jung (CERN-2005-014, Geneva, 2005), p.~575,
  %``The Les Houches Accord PDFs (LHAPDF) and Lhaglue,''
  hep-ph/0508110.

\bibitem{Denner:2005fg}
  A.~Denner, S.~Dittmaier, M.~Roth and L.~H.~Wieders,
  %``Electroweak corrections to charged-current e+ e- $\to$ 4 fermion processes:
  %Technical details and further results,''
  Nucl.\ Phys.\  B {\bf 724} (2005) 247
  [hep-ph/0505042];
%\bibitem{Denner:2006ic}
  A.~Denner and S.~Dittmaier,
  % ``The complex-mass scheme for perturbative calculations with unstable
  %particles,''
  Nucl.\ Phys.\ Proc.\ Suppl.\  {\bf 160} (2006) 22
  [hep-ph/0605312].
  %%CITATION = HEP-PH 0505042;%%




%\cite{Denner:2000bj}
\bibitem{Denner:2000bj}
  A.~Denner, S.~Dittmaier, M.~Roth and D.~Wackeroth,
  %``Electroweak radiative corrections to e+ e- ---> W W ---> 4 fermions in double pole approximation: The RACOONWW approach,''
  Nucl.\ Phys.\ B {\bf 587} (2000) 67
  [hep-ph/0006307].
  %%CITATION = HEP-PH/0006307;%%
  %146 citations counted in INSPIRE as of 02 Aug 2013
\bibitem{Kublbeck:1990xc}
  J.~K\"ublbeck, M.~B\"ohm and A.~Denner,
  %``Feyn Arts: Computer Algebraic Generation Of Feynman Graphs And Amplitudes,''
  Comput.\ Phys.\ Commun.\  {\bf 60} (1990) 165;\\
  %%CITATION = CPHCB,60,165;%%
  H.~Eck and J.~K\"ublbeck, {\it Guide to FeynArts 1.0\/},
  University of W\"urzburg, 1992.
\bibitem{Hahn:2000kx}
  T.~Hahn,
  %``Generating Feynman diagrams and amplitudes with FeynArts 3,''
  Comput.\ Phys.\ Commun.\  {\bf 140} (2001) 418
  [hep-ph/0012260].
\bibitem{Hahn:1998yk}
  T.~Hahn and M.~P\'erez-Victoria,
  %``Automatized one-loop calculations in four and D dimensions,''
  Comput.\ Phys.\ Commun.\  {\bf 118} (1999) 153
  [hep-ph/9807565].
%\cite{Hahn:2001rv}
\bibitem{Hahn:2001rv}
  T.~Hahn, C.~Schappacher,
  % ``The Implementation of the minimal supersymmetric standard model in
  % FeynArts and FormCalc,''
  Comput.\ Phys.\ Commun.\  {\bf 143 } (2002)  54-68.
  [hep-ph/0105349].
\bibitem{vanOldenborgh:1989wn}
  G.~J.~van Oldenborgh, J.~A.~M.~Vermaseren,
  %``New Algorithms for One Loop Integrals,''
  Z.\ Phys.\  {\bf C46 } (1990)  425-438.
%\cite{Hamilton:2006xz}
%\cite{Alwall:2007st}
\bibitem{Alwall:2007st}
  J.~Alwall {\it et al.},
  %``MadGraph/MadEvent v4: The New Web Generation,''
  JHEP {\bf 0709 } (2007)  028.
  [arXiv:0706.2334 [hep-ph]].

\bibitem{Hamilton:2006xz}
  K.~Hamilton and P.~Richardson,
  %``Simulation of QED radiation in particle decays using the YFS formalism,''
  JHEP {\bf 0607} (2006) 010
  [hep-ph/0603034].
  %%CITATION = HEP-PH/0603034;%%
  %20 citations counted in INSPIRE as of 02 Aug 2013
%\cite{Yennie:1961ad}
\bibitem{Yennie:1961ad}
  D.~R.~Yennie, S.~C.~Frautschi and H.~Suura,
  %``The infrared divergence phenomena and high-energy processes,''
  Annals Phys.\  {\bf 13} (1961) 379.
  %%CITATION = APNYA,13,379;%%
  %934 citations counted in INSPIRE as of 02 Aug 2013

%\cite{Falgari:2013gwa}
\bibitem{Falgari:2013gwa}
  P.~Falgari, A.~S.~Papanastasiou and A.~Signer,
  %``Finite-width effects in unstable-particle production at hadron colliders,''
  JHEP {\bf 1305} (2013) 156
  [arXiv:1303.5299 [hep-ph]].
  %%CITATION = ARXIV:1303.5299;%%
  %5 citations counted in INSPIRE as of 18 Dec 2013

%\cite{Rubin:2010xp}
\bibitem{Rubin:2010xp}
  M.~Rubin, G.~P.~Salam and S.~Sapeta,
  %``Giant QCD K-factors beyond NLO,''
  JHEP {\bf 1009} (2010) 084
  [arXiv:1006.2144 [hep-ph]].
  %%CITATION = ARXIV:1006.2144;%%
  %43 citations counted in INSPIRE as of 02 Aug 2013





%\cite{Frixione:2007vw}
\bibitem{Frixione:2007vw}
  S.~Frixione, P.~Nason and C.~Oleari,
  %``Matching NLO QCD computations with Parton Shower simulations: the POWHEG method,''
  JHEP {\bf 0711} (2007) 070
  [arXiv:0709.2092 [hep-ph]].
  %%CITATION = ARXIV:0709.2092;%%
  %760 citations counted in INSPIRE as of 21 Oct 2013



%\cite{Buckley:2011ms}
\bibitem{Buckley:2011ms}
  A.~Buckley, J.~Butterworth, S.~Gieseke, D.~Grellscheid, S.~H{\"o}che, H.~Hoeth, F.~Krauss and L.~L{\"o}nnblad {\it et al.},
  %``General-purpose event generators for LHC physics,''
  Phys.\ Rept.\  {\bf 504} (2011) 145
  [arXiv:1101.2599 [hep-ph]].
  %%CITATION = ARXIV:1101.2599;%%
  %114 citations counted in INSPIRE as of 21 Oct 2013
%\cite{Gieseke:2013eva}

\bibitem{Gieseke:2013eva}
  S.~Gieseke,
  %``Simulation of jets at colliders,''
  Prog.\ Part.\ Nucl.\ Phys.\  {\bf 72} (2013) 155.
  %%CITATION = PPNPD,72,155;%%


% %\cite{Catani:1996vz}
% \bibitem{Catani:1996vz}
%   S.~Catani and M.~H.~Seymour,
%   %``A General algorithm for calculating jet cross-sections in NLO QCD,''
%   Nucl.\ Phys.\ B {\bf 485} (1997) 291
%    [Erratum-ibid.\ B {\bf 510} (1998) 503]
%   [hep-ph/9605323].
%   %%CITATION = HEP-PH/9605323;%%
%   %976 citations counted in INSPIRE as of 20 Aug 2013
% %\cite{Kinoshita:1962ur}
% \bibitem{Kinoshita:1962ur}
%   T.~Kinoshita,
%   %``Mass singularities of Feynman amplitudes,''
%   J.\ Math.\ Phys.\  {\bf 3} (1962) 650.
%   %%CITATION = JMAPA,3,650;%%
%   %1050 citations counted in INSPIRE as of 20 Aug 2013
% %\cite{Lee:1964is}
% \bibitem{Lee:1964is}
%   T.~D.~Lee and M.~Nauenberg,
%   %``Degenerate Systems and Mass Singularities,''
%   Phys.\ Rev.\  {\bf 133} (1964) B1549.
%   %%CITATION = PHRVA,133,B1549;%%
%   %911 citations counted in INSPIRE as of 20 Aug 2013


\end{thebibliography}
